Title: Intramedullary spinal cord metastasis from colonic carcinoma presenting as Brown-Sequard syndrome: a case report

Authors:

Mohammed A Kaballo (mak9mak9@hotmail.com)
Darren D Brennan (darrenbrennanmd@gmail.com)
Mazen El Bassiouni (melbassiouni@materprivate.ie)
Stephen J Skehan (skehan67@eircom.net)
Rajnish K Gupta (rajnish.gupta@hse.ie)

Version: 3 Date: 10 May 2011

Author's response to reviews: see over
9th May 2011

Dear JMCR Editorial Team,

Re: Intramedullary spinal cord metastasis from colonic carcinoma presenting as Brown-Sequard syndrome: a case report

We would like to thank you for reviewing our manuscript. We have addressed your comments as follows –

1. The title was changed to: Intramedullary spinal cord metastasis from colonic carcinoma presenting as Brown-Sequard syndrome: a case report

2. All dates were removed to protect the patient’s identity.

3. An authors’ contributions section was added to the manuscript.

Thank you

Kind regards,

Yours sincerely,